

Spatiotemporal characteristics and earthquake statistics of the 2020 and 2022 adjacent earthquake sequences in North Aegean Sea (Greece)

P. Bonatis, * 10¹, V. Karakostas, 10¹, C. Kourouklas, 10¹, A. Kostoglou, 10^{1,2}, E. Papadimitriou 10¹

¹Geophysics Department, Faculty of Geology, Aristotle University of Thessaloniki, 54124 Thessaloniki, Greece, ²Institute of Geophysics, Polish Academy of Sciences, Department of Seismology, Ks. Janusza 64 Street, 01-452 Warsaw, Poland

Author contributions: Conceptualization: P. Bonatis, V. Karakostas, A. Kostoglou. Data Curation: P. Bonatis, V. Karakostas, C. Kourouklas. Formal Analysis: P. Bonatis, C. Kourouklas, A. Kostoglou. Investigation: All authors. Methodology: P. Bonatis, V. Karakostas, C. Kourouklas. Resources: V. Karakostas, E. Papadimitriou. Software: P. Bonatis, V. Karakostas, C. Kourouklas, A. Kostoglou. Supervision: V. Karakostas, E. Papadimitriou. Visualization: P. Bonatis, C. Kourouklas. Writing – original draft: P. Bonatis, C. Kourouklas, A. Kostoglou, E. Papadimitriou. Writing – review & editing: All authors.

Abstract The two moderate earthquakes that occurred close and to the north of the North Aegean Trough (NAT) on 26 September 2020 (Mw5.3) and 16 January 2022 (Mw5.4), both followed by aftershock activity, are examined. Seismic activity along the NAT and its parallel branches is continuous and remarkable, with numerous strong instrumental (M \geq 6.0) earthquakes. Yet, the frequency of moderate (5.0 \leq M<6.0) earthquakes outside these major fault branches is rather rare and therefore their investigation provides the optimal means to decipher the seismotectonic properties of the broader area. The temporal and spatial proximity of the two seismic activity with the employment of earthquake relocation techniques, moment tensor solutions and statistical analysis. Our research revealed that this seismic activity purely falls inside the Mainshock – Aftershock type, with fast aftershock decay rates and moderate productivity. According to our findings, the two seismic sequences, despite their close proximity, exhibit distinctive features as a result of the intricate stress field generated at the western termination of the NAF system in an extensional domain.

Περίληψη Αντικείμενο της μελέτης μας είναι οι δύο σεισμοί ενδιαμέσου μεγέθους που έγιναν πλησίον και βόρεια της τάφρου του Β. Αιγαίου στις 26 Σεπτεμβρίου 2020 (Mw5.3) και στις 16 Ιανουαρίου 2022 (Mw5.4) μαζί με τις μετασεισμικές τους ακολουθίες. Η σεισμική δραστηριότητα κατά μήκος της τάφρου του Β. Αιγαίου και των παράλληλων κλάδων της δεν είναι σπάνια, με πολλούς ισχυρούς σεισμούς (Μ≥6.0) να έχουν καταγραφεί κατά την ενόργανη περίοδο της σεισμικότητας. Από την άλλη πλευρά, ενδιαμέσου μεγέθους (5.0≤M<6.0) σεισμοί εκτός αυτών των κύριων κλάδων συμβαίνουν πολύ σπάνια, ως εκ τούτου η μελέτη τους αποτελεί μία πρώτης τάξης ευκαιρία για την αναλυτική ερμηνεία των σεισμοτεκτονικών ιδιοτήτων της ευρύτερης περιοχής. Η χωρική και χρονική εγγύτητα των δύο σεισμικών εξάρσεων που τέθηκε σε εξέλιξη από τα τέλη Σεπτεμβρίου του 2022 έως τις αρχές του 2022 ευνοεί την ενδελεχή διερεύνηση της σεισμικής δραστηριότητας με τη χρήση τεχνικών για τον επαναπροσδιορισμό των εστιακών συντεταγμένων, τον καθορισμό των μηχανισμών γένεσης τους και στατιστική ανάλυση της σεισμικότητας. Τα αποτελέσματα της στατιστικής ανάλυσης που πραγματοποιήθηκε υποδεικνύει ότι οι σεισμικές αυτές εξάρσεις ακολουθούν τυπικό μοτίβο ακολουθίας Κύριος σεισμός-Μετασεισμοί, με υψηλή απόσβεση των ρυθμών σεισμικότητας και μέτριου βαθμού παραγωγικότητα. Σύμφωνα με τα ευρήματά μας οι δύο σεισμικές ακολουθίες, παρά τη χωροχρονική τους εγγύτητα παρουσιάζουν διακριτά χαρακτηριστικά ως αποτέλεσμα του περίπλοκου τοπικού καθεστώτος τάσεων που οφείλεται στον τερματισμό της ζώνης μετασχηματισμού της Β. Ανατολίας σε ένα εφελκυστικό καθεστώς.

Non-technical summary On 26 September 2020 and 16 January 2022 two moderate earthquakes (M_w5.3 and M_w5.4, respectively) occurred at the North Aegean Sea, southern of Chalkidiki peninsula. Their close proximity in space and time and the rare manifestation of such moderate events in the area promotes their analysis in order to better understand the faults and the state of stress in the broader area. We used the available seismological stations in the area to enhance the quality of the earthquake catalog and thoroughly investigated the properties of the two seismic sequences. We found that this activity is not directly related to the prevailing seismotectonic feature of the area, namely the North Aegean Trough, but it is derived from secondary features, typically found in the vicinity of such large complex systems.

Froduction Editor: Gareth Funning Handling Editor: Yen Joe Tan Copy & Layout Editor: Jack B. Muir

Signed reviewer(s): Kostas Leptokaropoulos

> Received: 2023-11-03 Accepted: 2024-03-19 Published: 2024-05-08

^{*}Corresponding author: mponatis@geo.auth.gr

1 Introduction

The rollback of the oceanic lithospheric plate of eastern Mediterranean that subducts beneath the continental crust of the Aegean microplate (Pichon and Angelier, 1979; Papazachos and Comninakis, 1971) is the driving mechanism for the fast extensional deformation in an almost N-S direction of the back arc region (Konstantinou et al., 2017; Kapetanidis and Kassaras, 2019). The westward prolongation of the North Anatolian Fault (NAF) into the Aegean forms the North Aegean Trough (NAT) that constitutes the active boundary between the Aegean microplate and the Eurasian lithospheric plate (inset of Figure 1). McKenzie (1972) showed that the northward motion of the Arabian plate pushes the smaller Anatolian plate that is moving westerly relative to the Eurasian plate along the NAF, with an average velocity of about ~24 mm/yr. An additional N-S deformation of ~11 mm/yr in the Aegean further enhances this motion, resulting in a total SW motion of ~41 mm/yr of the south Aegean relative to Eurasia. More recent studies (e.g. England et al., 2016; Bitharis et al., 2023) comply with this overall pattern. A large part of the deformation occurs seismically, as the Aegean shows a total seismic slip rate of the order of 20 mm/yr relative to Eurasia (Papazachos and Kiratzi, 1996). Seismicity is intense and continuous in the back arc region forming specific seismic zones striking almost E-W in the normal faulting environment and ENE-WSW to NE-SW for the dextral strike-slip and oblique f).aulting seismic zones (Papazachos et al., 1998). Conjugate faulting sinistral strike-slip faults are also present (Karakostas et al., 2003) accommodating less frequent M≥6.0 main shocks as well as moderate ($M \ge 5.0$) earthquakes.

Several strong (M≥6.0) historical and instrumental earthquakes have struck the NAT (Figure 1) since 360 BC (Papazachos and Papazachou, 2003). Their temporal distribution evidences catalog incompleteness at least until 1300 AD, and afterward missing events are the ones of M<6.5 (Kourouklas et al., 2018). Most recently, the 2014 M6.9 main shock ruptured the middle part of the NAT (Kiratzi et al., 2016), taking place in a previously identified seismic gap (Karakostas et al., 1986). The aftershock activity was astonishingly weak with a maximum magnitude aftershock of M4.9. However, the western termination of NAT against the Greek mainland appears to be relatively quiescent (Kourouklas et al., 2018, 2022). The strike-slip faulting of the NAT is confirmed by the fault plane solutions of the stronger earthquakes in the most recent part of the instrumental era, when seismological networks were capable to provide adequate data, such as in the case of the 1983 seismic sequence (Rocca et al., 1985). The results from waveform modeling performed for the 1968 (M7.5; Pacheco and Sykes (1992)), 1981 (M6.8) and 1983 (M6.6) earthquakes established the dextral strike-slip nature of the major faults in North Aegean with the axis of maximum extension, T, striking N–S and being nearly horizontal (Kiratzi et al., 1991).

Because it is difficult to determine the causative faults of the moderate (M \geq 5.0) earthquakes, our understanding of the seismogenic setting of the secondary faults in

our study area is rather limited. An M5.8 main shock occurred in 2013 on one of the ENE–WSW trending parallel dextral strike-slip fault branches in Northern Aegean, in the continuation of the 1968 large (M=7.5) rupture (Drakopoulos and Ekonomides, 1972). Its rich aftershock sequence was scrutinized and the off–fault seismicity was perfectly explained with Coulomb stress changes when the parameters of friction and Skempton's coefficients attained values of μ >0.5 and B=0.0, respectively, implying high fault friction (Karakostas et al., 2014) Thus, clarifying the properties of the causative faults of moderate earthquakes is crucial to reveal the seismogenic structures and their relation to the major fault zones and evaluate the subsequent seismic hazard.

Microseismicity studies are not common in the study area, given that active faults are off-shore, and nearfault instruments are not in operation. The western termination of NAT was investigated by Hatzfeld et al. (1999) who confirmed the strike-slip motion which has probably been active since the Pliocene. This strike-slip motion is transferred into normal faulting (with the direction of principal extension being constant) in continental Greece. Accurate relocation of seismicity during 2011–2016 along the NAT yielded a pick of the focal depth distribution at 8 km diminishing down to 20 km (Konstantinou, 2017). The relocated seismicity defines distinctive clusters, one of which is located inside our study area, but the spatial distribution is quite disperse and thus not adequate to identify a specific activated structure.

On 26 September 2020 (at 22:50:24 UTC), an M_w5.3 earthquake occurred (right magenta circle in Figure 1), to the north of the NE-SW trending western portion of NAT and very close to its inferred trace, in a way that it would be considered as associated with failure on a fault patch along the NAT interface. It occurred in the middle of the night and was widely felt, alarming plenty of citizens even at distances greater than 200 km (EMSC felt reports). The GCMT (Global Centroid Moment Tensor; https://www.globalcmt.org/) solution suggests strike-slip faulting with one of the nodal planes striking NE-SW that complies with the sense of motion along the NAT (focal mechanisms configured with magenta compressional quadrants in Figure 1). The appreciable aftershock activity, however, persistently aligned in a NW-SE direction in agreement with the strike of the second nodal plane. These first observations provoked our interest in investigating the characteristics of the activated structure. Then, fifteen months later, a second moderate M_w5.4 earthquake occurred on 16 January 2022 (at 11:48:06 UTC) very close to the epicenter of the first 2020 shock (left magenta circle in Figure 1). The 2022 aftershock activity exhibited the same pattern as in 2020, and the second main shock's GCMT solution implies oblique faulting with the one nodal plane also striking NW-SE. In this study, we relocate the aftershocks and determine the fault plane solutions of the strongest among them. We further study the spatiotemporal evolution of the activity and perform statistical analysis through the application of the Epidemic Type Aftershock Sequence (ETAS) model.



Figure 1 Epicentral distribution of M≥4.1 crustal (0≤h≤20 km) earthquakes occurred in northern Aegean Sea from 1975 to 2022 (after Leptokaropoulos et al., 2012), as compiled from the regional parametric earthquake catalog of the Seismological Station of Geophysics Department of the Aristotle University of Thessaloniki (http://geophysics.geo.auth.gr/ss/catalogs_en.html; Aristotle University of Thessaloniki, 1981). The small light green and moderate orange circles depict the epicenters of 4.1≤M<5.0 and 5.0≤M<6.0 earthquakes, respectively, and magenta circles the epicenters of for the earthquakes investigated in this study. White stars display the epicenters of the historical M≥6.5 earthquakes between 1845–1900. Yellow stars depict epicenters of the M≥6.0 earthquakes occurred during the instrumental era. The available fault plane solutions of the M≥5.0 earthquakes as taken from Global Centroid Moment Tensor (GCMT; https://www.globalcmt.org/) are plotted as equal area lower hemisphere projections with the compressional quadrants colored according to their magnitude as before. Solid thick red line represents the main branch of the North Aegean Trough Fault Zone, whereas the parallel red arrows imply the right-lateral strike-slip motion. Inset map shows the main seismotectonic features of the Aegean region (solid red lines; KTFZ: Kefalonia Transform Fault Zone; RTF: Rodos Transform Fault; NAF: North Anatolian Fault).

2 Seismological Data and Methods

2.1 Earthquake Relocation

An initial earthquake catalog was compiled by retrieving the recordings of the Hellenic Unified Seismological Network (HUSN) after routine analysis accomplished in the central Seismological Station of the Geophysics Department of the Aristotle University of Thessaloniki (http://geophysics.geo.auth.gr/ss/) for earthquakes with M≥2.5. Additional phase picking and initial location were performed by the authors of the present study to achieve the inclusion of lower magnitude events (even lower than M=1.0) that were detected by stations quite close to the activated area. In total, 841 earthquakes were initially located with magnitudes 0.6<M<5.4. For the earthquake relocation procedure, we used data from 14 seismological stations at epicentral distances up to approximately 150 km (Figures S1 & S2). The median azimuthal gap for all initial locations equals to 108° indicating a sufficient station azimuthal coverage, considering that the activated structures are offshore.

The first step towards improved focal coordinate estimation is relocation with the HYPOINVERSE code (Klein, 2002). This step requires the manually picked Pand S- phases, a local velocity model, the ratio of compressional to shear wave velocity (V_p/V_s) and the corresponding station time delays. At first, we estimated the V_p/V_s through the Wadati method (Wadati, 1933) by taking the minimum number of phase arrival pairs (P- and S-) equal to 8 for each event, and the process resulted in a ratio equal to 1.74 ± 0.007. Similar values were also estimated for the broader Aegean region in previous studies (e.g. Karamanos et al., 2007; Mesimeri et al., 2018; Andinisari et al., 2020; Karakostas et al., 2021). Next, we proceeded to the determination of a crustal model using the VELEST algorithm (Kissling et al., 1994) by testing multiple published models as reference ones (e.g. Akyol et al., 2006; Karabulut et al., 2006; Konstantinou, 2018). All resulting models after multiple iterations consistently converged to a similar model (apart from the first few kilometers where differences could be found) very close to the model of Karabulut et al. (2006). We thus decided to adopt this model as a reference one (Figure S3) and use our derived model (Table 1) for the relocation procedure. We also added the appropriate station corrections calculated with the VELEST application to incorporate lateral inhomogeneities in the 1-D crustal model.

We then run the hypoDD program (Waldhauser, 2001) that employs the double difference algorithm (Waldhauser and Ellsworth, 2000), to improve the accuracy of the focal coordinates as they were obtained from HY-POINVERSE. Travel time differences between manually picked phases in the earthquake catalog were calculated, and then we kept event pairs with at least eight (8) observations, resulting to 32,956 P-phase pairs and 25,870 S-phase pairs (13 links per pair on average) in total. The differential times data set was analyzed using five sets with five iterations on each set gradually decreasing the residual threshold (secs) and the maximum distance (km) between catalog linked pairs. The hori-

Depth (km)	V _p (km/s)
0.0-2.0	3.56
2.0-4.0	4.32
4.0-7.0	5.43
7.0-14.0	6.08
14.0-20.0	6.22
20.0-26.0	6.55
26.0-29.0	6.8
≥30	7.29

Table 1P-wave velocity (V_p) model adjusted for the study
area

zontal and vertical errors between the initial and the relocated catalog indicate a substantial reduction of location uncertainty (Figure S4). The final catalog contains 817 relocated earthquakes out of 841 (~97%) that constituted the initial data set. The 477 earthquakes occurred before the 2022 main shock and the 340 are the aftershocks of the second sequence.

2.2 Fault plane solutions

Fault plane solutions of 12 earthquakes with $M \ge 4.0$, which occurred during the whole study period, were calculated. One of them occurred slightly later, in April 2022, but in the same area. The moment tensor inversions were conducted using the Grond software (Heimann et al., 2018) which operates under the Pyrocko toolbox framework (Heimann et al., 2017). The methodology applied within Grond aims to minimize the misfit between synthetic and observed data by implementing a Bayesian bootstrapping inversion in parallel bootstrap chains. In the present case, the bootstrap algorithm was applied to minimize the L2 norm misfit for 22000 iterations, 3000 to uniformly sample the solution space and 19000 for the direct sampling. The number of parallel bootstrap chains was set equal to 200. We used the recordings of the regional broadband seismological stations of HUSN in distances ranging between 50 and 300 km (Figure S1). Green's functions were calculated for the local velocity model (Table 1) using the QSEIS program (Wang, 1999). A point source model was considered to perform the Bayesian optimization for a deviatoric moment tensor. Lastly, bandpass filters were applied, using frequencies between 0.04 and 0.11 Hz (the corresponding window for each earthquake is presented in Table 2) and the data fitting was carried out in the time domain. The NW-SE striking nodal planes of the focal mechanisms (Table 2; Table S1) exhibit predominantly sinistral strike-slip sense of motion, ranging from almost pure strike-slip to oblique faulting.

No	Date	Origin time	Lat	Long	Depth	Mar	Freq	Strike	Dip	Rake	Misfit ²
			(°)	(°)	(km)	1*100	(Hz)	(°)	(°)	(°) (°) (°instru	
1	2020/09/26	18:39:20	39.959	24.308	21.3	4.3	0.04-0.09	334	86	09	0.135
2	2020/09/26	22:50:25	39.954	24.306	18.9	5.3	0.04-0.08	145	73	-18	0.132
3	2020/09/27	12:22:20	39.983	24.315	11.5	4.4	0.04-0.09	131	51	-42	0.106
4	2020/09/27	12:58:02	39.943	24.320	15.6	4.3	0.04-0.09	339	85	-44	0.175
5	2020/09/27	16:05:08	39.952	24.329	19.1	4.1	0.05-0.10	310	53	-71	0.164
6	2020/09/28	04:12:42	39.943	24.296	18.9	4.7	0.04-0.08	147	76	-07	0.095
7	2022/01/16	11:48:06	39.980	24.292	17.6	5.4	0.04-0.08	125	52	-43	0.100
8	2022/01/16	12:26:19	39.999	24.276	12.7	4.5	0.05-0.09	185	68	18	0.109
9	2022/01/16	13:36:58	39.990	24.302	11.2	4.0	0.05-0.09	132	63	-33	0.144
10	2022/01/16	18:29:27	39.977	24.314	11.7	4.1	0.06-0.1	135	66	-28	0.111
11	2022/01/16	22:32:00	39.989	24.317	12.4	4.3	0.05-0.09	128	65	-31	0.091
12	2022/04/20	00:29:03	39.934	24.331	8.0	4.0	0.06-0.11	336	68	-30	0.124

SEISMICA | RESEARCH ARTICLE | Characteristics and statistics of the 2020 and 2022 North Aegean Sea earthquake sequences

Table 2 Fault plane solutions calculated for eleven earthquakes of the relocated catalog and one that occurred one month later (No 12). The focal parameters provided correspond to the relocated ones. In the last column, the square of the misfit of each solution is reported.

2.3 The Epidemic-Type Aftershock Sequence (ETAS) model

Short-term temporal properties of both the 2020 and 2022 seismic excitations were studied through the ETAS stochastic model (Ogata, 1988, 1998). The temporal ETAS model (Ogata, 1988) expresses the seismicity occurrence rate density as the summation of two parts, the constant background seismicity rate, μ , assumed time-independent, and the occurrence density rate of triggered earthquakes, $\lambda_i(t)$, and is given by a conditional intensity function, $\lambda(t)$:

$$\lambda(t) = \mu + \lambda_i(t) = \mu + \sum_{[i, t_i < t]} \frac{K e^{\alpha(m_i - m_c)}}{(c + t - t_i)^p}, \quad (1)$$

where K and α are the aftershock productivity parameters, c and p are the parameters of the temporal aftershock decay rate of the modified Omori law (Utsu, 1961), m_i is the magnitude of each earthquake occurred at time t_i and m_c is the completeness magnitude. The parameter K represents the intensity of aftershocks generation above m_c triggered by an earthquake with m = m_c , whereas α parameter describes the efficiency of earthquakes in triggering their own aftershocks. Large α values [1.3-3.1] indicate that large magnitude earthquakes trigger a large number of aftershocks, whereas small α [0.35-0.85] implies relatively higher triggering capabilities of small earthquakes. This means that Mainshock-Aftershock sequences typically tend to have larger α values, dominated by the main shock magnitude, while a swarm-like activity is characterized by small α values (Hainzl and Ogata, 2005; Ogata, 1992). The exponent p of the modified Omori law controls the aftershocks decay rate, where the decay is becoming faster as the value of p is increasing. The parameter cis linked with the short-term aftershock incompleteness soon after the occurrence of a main shock (t = 0), aiming to avoid singularities at occurrence times very close to t = 0. The five parameters of the model were estimated via the Maximum Likelihood Estimation (MLE) method using the simulated annealing technique proposed by Lombardi (2015) and implemented through an algorithm which is part of the SEDA package (Lombardi, 2017).

For the model evaluation, a residual analysis was performed Ogata (1988, 1998), which offers a qualitative model evaluation through visual display. Specifically, the occurrence times, t_i , were converted into transformed times, τ_i , according to the function:

$$\tau_i = \int_0^{t_i} \lambda(t) dt. \tag{2}$$

Transformed times, τ_i , express the number of earthquakes that are expected to occur in the time interval $[0, t_i]$. If the estimated model adequately describes the temporal seismicity evolution, then transformed data, so-called residuals, behave like a stationary unit rate Poisson process. Otherwise, the transformed process will show some systematic departure from the linear Poisson process. Positive and negative departures indicate that the estimated model is under- and overpredicting the observed seismicity, respectively. In order to evaluate whether the transformed times, τ_i , are described by the Poisson process, the one-sample Kolmogorov-Smirnov test (KS1 test; Massey (1951)) was applied. More specifically, if τ_i are modelled by the unit rate Poisson process, then the transformed earthquake interevent times, $t_{i+1} - t_i$, should be independent and identically drawn from an exponential distribution (Llenos and Michael, 2013). The KS1 test is implemented under the null hypothesis that the earthquake interevent times are following the exponential distribution based on the p-value returned by the test, compared with the 0.05 significance level. If p-value is greater (or lower) than 0.05 then the null hypothesis can either be rejected or accepted.

3 Results

3.1 Spatiotemporal evolution of seismicity

The number of precisely relocated earthquakes equals to 817, with 477 of them belonging to the period starting from the initiation of the seismic excitation (26 September 2020) until the start of the second seismic sequence (16 January 2022) and the rest 340 constituting the second aftershock sequence. Regarding pre-seismic activity, only two (2) foreshocks were detected prior to the first main shock with the largest one occurring 4 hours in advance (Table 2). Our interpretations on aftershock activity and the associated active faults can be ensured by the quality criteria that the relocation procedure has fulfilled concerning the uncertainty estimation. These criteria do not prevent from systematic bias, but with our efforts to approach the velocity structure as best as possible we are confident that this bias could be considered small.

We are interested in how seismicity evolved spatially or temporally in the activated area on the temporal scale of several months, for which that this activity persisted. We started our relocated data set since 1 January 2020, almost 10 months before the first main shock occurrence. We may observe that seismicity is quite sparse with a couple of 2.0<M<2.9 shocks located close to the first main shock epicenter (green dots since the beginning of the space-time plot up to 26 September 2020, in Figure 2a) but not close in time for being considered foreshock activity as part of the nucleation phase. An M_w4.3 earthquake, instead, occurred about 4 hours before and in close distance with the main shock (Table 2). Intense aftershock activity followed the main shock on 26 September 2020, extended in an area more than 15 km long, much larger than the rupture length of an M_w5.3 main shock, (roughly 6 km) as prescribed by empirical scaling laws (Wells and Coppersmith, 1994; Thingbaijam et al., 2017).



SEISMICA | RESEARCH ARTICLE | Characteristics and statistics of the 2020 and 2022 North Aegean Sea earthquake sequences

Figure 2 (a) Spatiotemporal distribution of the relocated seismicity since 1 January 2019 up to 1 November 2022. Yellow stars represent the two main shocks, red circles the $4.0 \le M_w \le 4.9$, orange circles the $3.0 \le M \le 3.9$, green circles the $2.0 \le M \le 2.9$, and small black dots the M ≤ 2.0 earthquakes. (b) Histogram of the focal depths with different colors denoting the two aftershock sets.

As Figures 2a & 3a show, seismicity is mainly concentrated on a 20 km length prolonged zone, with a general strike of 145° (thick dashed line in Figure 3a). The epicentral distribution of the two distinct clusters present a spatial overlap of slightly over 50%. The determined fault plane solutions unveil the dominance of strike-slip faulting in the study area given that pure strike-slip (e.g., No 1 & 6, Table 2; Figure 3a) and strike-slip with a normal component (No 2 & 4 & 9-12) focal mechanisms are present. The complex faulting patterns coming into play are revealed by the presence of either pure normal (No 5, Table 2; Figure 3a) or oblique (normal with a strike-slip component) focal mechanisms (No 3 & 7, Table 2; Figure 3a).

In Figures 4 & 5 the relocated seismicity of the two activated structures is shown separately aiming to detail the properties of each seismic sequence. Regarding the 2020 M_w 5.3 sequence (Figure 4a), the epicentral alignment (NW-SE) of the aftershocks is in good agreement with the strike of the one nodal plane of the main shock focal mechanism. The total length of the activated area is approximately 15 km, which significantly exceeds the estimations of frequently used empirical relationships between main shock magnitude and the causative fault length (Wells and Coppersmith, 1994; Thingbaijam et al., 2017). The duration of the intense aftershock activity was short, with over half of the aftershocks occurring within the first week after the main shock (Figure 3). Moreover, all M≥4.0 aftershocks took place in less than 3 days (Table 2). An interesting feature of this seismic sequence is also the occurrence of a few foreshocks with the strongest among them having a magnitude of M_w4.4 (Table 2), placed very close to the main shock epicenter (Figure 2) although in deeper parts of the seismogenic layer.

The aftershock activity during the first 2 days defines a seismogenic zone of 9 km (13 – 22 km) (Figure 4b). As time advances, however, the inclusion of all earthquakes (Figure 4c) evidences an expansion of the aftershock zone in more shallow depths (~5km). Normal to the strike cross section encompassing events within the larger part of the aftershock distribution (3 km of either side of the cross section) provides a complete picture of the activated fault plane (Figure 4d). The total depth extent of the aftershock zone ranges from 6 to 24 km (Figure 2b).

Shifting our focus to the second seismic sequence that emerged from the strongest event included in our catalogue (M_w5.4) at the beginning of 2022 we might observe that the epicentral alignment of the aftershocks is analogous to the previous case (NW-SE) however their spatial extent is developed further to the north. The total length of the activated area, as inferred by the aftershock epicentral distribution is approximately equal to 10 km (Figures 2a, 5a). The aftershock activity of the first 2 days occupies a depth range of 9 to 21 km, inside a 12 km extent (Figure 5b), with only minimal lowmagnitude events outlying. Figure 5c, which includes all events at a distance of 1.5 km either side of the normal cross section, appears to be very similar to the one of Figure 4b, highlighting the fast decay of the aftershock rate after the first few days. Moreover, the wider



Figure 3 Seismicity of the study area, since the beginning of 2019 until the end of 2022. (a) Stars indicate the relocated epicenters of the main shocks, whereas circles denote either the relocated aftershock locations or the background seismicity. All epicenters are scaled in compliance with the corresponding earthquake magnitudes and colorcoded according to the temporal scale at the bottom of the figure. Fault plane solutions determined in this study (Table 2) are also shown as lower hemisphere equal area projections, with the compressional quadrants colored in blue. (b) Temporal evolution of seismicity for the 2019–2022 period versus magnitude. Blue, yellow and cyan circles and shaded areas indicate the pre-2020, 2020 and 2022 seismic activity. Red solid line depicts the cumulative number of earthquakes for the same period.



Figure 4 a) Spatial distribution of the relocated aftershocks of the $M_w 5.3$ main shock on 26 September 2020 (red star). The epicenters of the aftershocks are shown as circles of different sizes and colors according to their magnitude. The black dashed line shows the general strike of the epicentral distribution, whereas the constant one signifies the normal to the strike cross sections including events b) within 1.5 km either side of the section, during the first 2 days after the main shock, c) within 1.5 km for the entire duration and d) 3.0 km either side of the section, again for the entire relocated catalog.

normal cross section (Figure 5d) reveals the close proximity of the strongest aftershocks (11-13 km in depth).

3.2 Temporal features from the ETAS model fitting

The temporal ETAS parameters (μ , K, α , c and p) were estimated through the MLE method after the determination of the magnitude of completeness, m_c , for the entire initial earthquake catalog covering the period 2019-2023. The m_c was identified through the Goodness-of-Fit method (GFT; Wiemer and Wyss (2000)), considering the 95% confidence level of residuals (Figure S5) and was found equal to m_c =1.9 (residuals value equal to 3.66%) resulting to a data set of 470 earthquakes with $m \geq m_c$ and a b-value equal to 0.78 (b=0.78). The model was first applied in the entire initial earthquake cata-



Figure 5 a) Spatial distribution of the relocated aftershocks of the $M_w 5.4$ main shock on 16^{th} January 2022 (red star). The epicenters of the aftershocks are plotted with circles of different sizes and colors according to their magnitude. The black dashed line shows the general strike of the epicentral distribution, whereas the constant one signifies the normal to the strike cross sections including events b) within 1.5 km either side of the section, during the first 2 days after the main shock, c) within 1.5 km for the entire duration and d) 3.0 km either side of the section, again for the entire relocated catalog.

Period	μ	K	α	c	p	Obs.	KS1 p-value
2019 -2022	0.046	0.015	1.78	0.030	1.23	470	0.62
01/01/2020 -30/6/2021	0.047	0.017	1.75	0.029	1.17	240	0.60
1/7/2021 -31/12/2022	0.037	0.011	1.86	0.037	1.25	197	0.57

Table 3 Temporal ETAS parameters estimates (μ , K, α , c and p) for the periods 2019-2022, January 2020-June 2021 and July 2021-December 2022, along with the respective number of observations and the p-values of the one sample Kolmogorov-Smirnov goodness-of-fit test (KS1) between the earthquake interevent times and the exponential distribution.

log of the period 2019-2022 and then in two additional and distinctive sub-periods, namely from January 2020 to June 2021 and from July 2021 to December 2022, aiming to compare the temporal properties of the two seismic sequences. The selection of each sub-period was based on the adequacy of the number of earthquakes before the occurrence of each excitation, in which the parameters were estimated, representing the learning phase of the ETAS model application. The calculated parameter values are given in Table 3.

The ETAS application for the entire period (2019-2022) ascertained that the model adjusts well the observed earthquake rate, with slight discrepancies soon after the occurrence of the 2020 M_w 5.3 and 2022 M_w =5.4 main shocks (Figure 6a). The good fit of the estimated model in respect to the observations is highlighted by the residuals analysis application (Figure 6b), since the lines that depict the expected and the observed number of events against the transformed times (red and black lines in Figure 6b, respectively) almost coincide. This latter fact is confirmed by the result of the KS1 goodness of fit test shown in Table 3. Specifically, the calculated p-value of the test is equal to 0.62, much larger than the 0.05 confidence level. The estimated parameters are acquiring values typical for Mainshock-Aftershock sequences (Ogata, 1992). In more detail, the observed seismicity during the period from 2019 to 2022 is characterized by a very low background rate equal to μ =0.046 event/day. This means that almost 88% of the total number of earthquakes are offsprings of the 2020 M_w 5.3 and 2022 M_w =5.4 sequences, and only 68 out of the 470 are assumed to be independent. Furthermore, the productivity parameter, α , was estimated equal to $\alpha = 1.78$, indicating also Mainshock-Aftershock type of sequence, since Mainshock-Aftershock activity is typically characterized by α values ranging between [1.3-3.1], whereas swarm-like activity is ascribed to values ranging between [0.35-0.85] (Hainzl and Ogata, 2005; Ogata, 1992). The parameters expressing temporal characteristics, c and *p*, attain expected values, in comparison with those reported in previous studies ranging from 0.03 to 0.07 and from 1.16 to 1.25, respectively (Chu et al., 2011; Kourouklas et al., 2020).

Placing emphasis on the two sub-periods (January 2020-June 2021 and July 2021-December 2022), when

the M_w5.3 and M_w5.4 main shocks occurred, the small discrepancies between the observed earthquake rates and the modeled ones are becoming more explicit (Figures 6c and 6e, for the January 2020-June 2021 and July 2021-December 2022, respectively). Specifically, it is observed that the estimated models for both periods slightly underestimate the earthquake rates soon after the occurrence of the $M_w 5.3$ and $M_w 5.4$ earthquakes. This result is also visible from the residual analysis plots (Figures 6d & 6f), in which the observed transformed times (red lines in Figures 6d & 6f) slightly deviate from the unit rate Poisson process (black lines in Figures 6d & 6f). However, the p-values of the KS1 test (p-value=0.60 and 0.57 for the periods January 2020-June 2021 and July 2021-December 2022, respectively; Table 3) are again much larger than the significance level (0.05), suggesting a good performance of the temporal ETAS model to the data from a statistical point of view.

Comparison of the parameter estimates of the two sub-periods (Table 3) shows that throughout the first one (January 2020-June 2021), during which the M_w5.3 earthquake occurred (September 26th of 2020), the background rate is estimated equal to $\mu = 0.047$ event/day, a value almost equal to the entire period estimate). On the contrary, background rate is found quite smaller $(\mu = 0.037 \text{ event/day})$ during the period from July 2021 to December 2022, in which the M_w5.4 earthquake occurred (January 16th of 2022). Both estimated values are again indicating a clear Mainshock-Aftershock activity as well as the estimated values of the productivity parameter, α , ($\alpha = 1.75$ and 1.86 for the 1st and the 2nd sub-periods, respectively; Table 3) with the one attached to the second sub-period being higher. Additionally, the estimated value of p parameter of the modified Omori law is larger in the second sub-period's application (p = 1.17 instead of p = 1.25 for the 1st period) indicating the smaller duration of the second sequence.



Figure 6 Observed (black lines) and expected by the estimated temporal ETAS model (red lines) cumulative earthquake number against ordinary (a,c,e) and transformed (b,d,f) time for the periods 2019 – 2022 (a & b, respectively), January 2020 – June 2021 (c & d, respectively) and July 2021 – December 2022 (e & f, respectively). Magnitudes of earthquakes with $m \ge m_c$ (right y-axis) against time are shown as pink dots in all subplots.

4 Discussion

As the westward prolongation of the North Anatolian Fault (NAF) system into the Aegean Sea, the North Aegean Trough (NAT) is characterized by dextral strikeslip faulting. The intense N-S extension of the back-arc area due to the roll back of the subducted slab in southern Aegean, resulted to a complex transtensional basin dated back to early Pliocene or Pleistocene. The conjunction of major strike-slip systems with a wide variety of secondary structures, especially at fault zone tips or between linked structures, the damage zones (e.g Petit and Barquins, 1988; Kim et al., 2004; Kim and Sanderson, 2006) is very commonly met. The reason behind their development can be attributed to stress concentrations (e.g Cox and Scholz, 1988) or to host displacement alterations along the fault zones (Kim et al., 2000). The North Aegean area is characterized by a broad spectrum of strike-slip damage patterns (Figure 7), with the most important and relevant to our study area being:

a) Horsetail structures: The western endpoint of the NAT is characterized by a right-stepping horsetail structure (red patch in Figure 7) expressed through numerous oblique splays stemming from the main strike-slip zone forming a number of transtensive basins (Sakellariou et al., 2017).

b) Positive or negative flower structures: Negative flower structures are found inside most of the subbasins bounded by the oblique splays. They consist of opposing-dip normal fault structures and merge into a single strike-slip fault at deeper parts (upper left part rectangle in Figure 7). Positive ones are also present but outside the area under study, along the south Marmara and the Ganos fault segments, expressing transpressive structures owing their formation to strike-slip faults with a reverse component (Rodriguez et al., 2023).

c) Conjugate faults: They are faults that intersect with the main structure at a high angle (typically more than 60°) and exhibit a sense of displacement opposite to the dominant one, with which they are often not connected. In our case, they represent the sinistral counterparts of the prevalent dextral strike-slip faults that dominate the northern Aegean area. Conjugate faults may be combined with branch faults (see below) and create block rotation (e.g. Nicholson et al., 1986; Kim et al., 2003).

d) Branch faults: They represent shear fractures having similar sense of motion as the main strike-slip zone (dextral strike-slip in our case, blue patch in Figure 7). They can act together with the main zone to create splays (e.g. Kim et al., 2004) or with other structures and form more complex patterns.

The diversity characterizing the termination of strikeslip fault systems, especially on such a large scale such as the NAT makes it very challenging to uncover its fine details, due to their complex nature and high intercorrelation. Typical examples are the fault segments associated with the two main shocks of this study which are not included in the simplified tectonic map of the North Aegean based on the interpretation of the air gun lithospheric profiles by Papanikolaou et al. (2006) as well as in other studies (e.g. Ferentinos et al., 2018; Sakellariou and Tsampouraki-Kraounaki, 2019).



Figure 7 Strike-slip damage patterns (schematic illustrations) and their connection to the study area (simplified map). The green and yellow stars indicate the epicenters of the main shocks of this study.

Based on our analysis, the activated structures hosting the 2020 M_w 5.3 and 2022 M_w 5.4 main shocks can be attributed to conjugate faults at the termination of the main strike-slip zone, having a considerable vertical component consistent with the extensional character of the NAF termination (Figure 7). The primary indications leading to this suggestion are the determined focal mechanisms of the main shocks and the stronger aftershocks (Table 2), which indicate an intricate stress field consisting of strike-slip and normal faulting style. Adding to this, the strike-slip moment tensor solutions exhibit a left-lateral displacement, as opposed to the dextral strike-slip motion characterizing the NAF. Moreover, the spatiotemporal distribution of aftershocks (Figures 3-5) further demonstrates the high angle (almost perpendicular) in which the activated structures lie in comparison to the NAF (Figure 1). The scenario of a transtensional flower structure cannot be ruled out given that such systems are closely related and common to conjugate fault systems. A valid reason for this argument can be ascribed to the relative differences of the focal depths of the two sequences (Figure 2). Even though the spatial distribution of both aftershock sequences appears to extend in more or less the same area in map view (Figure 3), the mean focal depths from the aftershocks originated from the 2020 M_w5.3 sinistral strike-slip main shock are found in deeper parts compared to those from the 2022 $M_w 5.4$ oblique (λ =-43°; normal with sinistral strike-slip component) one. In this case, a depressed area, or even a pull-apart basin (on a larger scale) is most commonly formed. Bathymetric maps (Papanikolaou et al. (2006) and references therein) cannot provide us with a definite answer however our study area is located at the northern margins of the main North Aegean basin and more specifically in an area where a bulge is disrupting the almost linear NE-SW oriented edges of the basin.

Statistical analysis of the short-term clustering features of the 2020-2022 excitations highlights their Mainshock-Aftershock nature. The estimated temporal ETAS parameters show non-significant background seismicity rate changes for both the average model (2019-2022) and the two sub-period models, which are referring to the 2020 and 2022 sequences, indicating that the excitation is driven by the regional tectonic loading. An interesting remark arises from the estimated parameters of the second sequence (2022) modeling, in which the temporal parameters (c and p) are larger. These values indicate a faster decay rate compared to the first sequence's aftershocks, even though the magnitude of the 2022 main shock is larger than the 2020 one. This remark could be likely explained by the fact that the study area was recently activated and consequently rather stress relaxed.

Moreover, the low background rate, the estimated productivity parameters and the fast aftershock decay rates in all evaluated models provide an indirect insight into the properties of the activated fault segments, qualifying them as rather weak. This property characterizing fault segments in the broader area (including the main branch of NAT fault zone) has already been suggested by physics-based earthquake simulator results (Kourouklas et al., 2021).

5 Conclusions

We thoroughly investigated the seismic sequences of two moderate main shocks occurred in the region north of NAT on 26 September 2020 (M_w 5.3) and 16 January 2022 (M_w 5.4), by means of their spatiotemporal distribution, moment tensor solutions and statistical analysis. We constrained the dimensions of the rupture areas (10x10 km² for the 2020 main shock and 11x10 km² for the 2022 main shock) based on the relocated early aftershocks. Fault plane solutions highlight the complexity of the faulting patterns, with most of them exhibiting mainly sinistral strike-slip faulting, with some oblique cases (strike-slip with a normal component) being also present.

All things considered, we attribute the occurrence of the investigated seismic sequences as a result of secondary faulting related to the termination of an active strike-slip plate boundary, expressed as transtensional deformation. We may consider these main shocks as a beneficial example of better recognition of the complex behavior characterizing the NAT, but through the lens of seismic hazard, we weigh this kind of events much less prominent than the overall seismic hazard of the broader area.

6 Acknowledgements

We would like to thank Kostas Leptokaropoulos and an anonymous reviewer for their valuable comments that helped to improve this manuscript. We also thank Yen Joe Tan for his editorial assistance. The SEDA software package (Lombardi, 2017) was used for the estimation of the temporal ETAS parameters (Available online: https://zenodo.org/record/55277; last accessed on March 2023). The software package of Pyrocko Heimann et al. (2017) used for the moment tensor inversions is available at http://pyrocko.org. Figures were created with the use of MATLAB software (Available online: http://mathworks.com; last accessed in March 2023) and Grapher version 16 (http://www.GoldenSoftware.com). The software Generic Mapping Tools (Wessel et al., 2013) was used to plot the maps and the cross sections. Geophysics Department Contribution 971.

7 Data and Code Availability

The seismic data used in this study are publicly available at https://doi.org/10.7914/SN/HT.

References

- Akyol, N., Zhu, L., Mitchell, B. J., Sözbilir, H., and Kekovalı, K. Crustal structure and local seismicity in western Anatolia. *Geophysical Journal International*, 166(3):1259–1269, Sept. 2006. doi: 10.1111/j.1365-246x.2006.03053.x.
- Andinisari, R., Konstantinou, K., and Ranjan, P. Seismotectonics of SE Aegean inferred from precise relative locations of shallow crustal earthquakes. *Journal of Seismology*, 24(1):1–22, 2020.
- Bitharis, S., Pikridas, C., Fotiou, A., and Rossikopoulos, D. GPS data analysis and geodetic velocity field investigation in Greece, 2001–2016. *GPS Solutions*, 28(1), Oct. 2023. doi: 10.1007/s10291-023-01549-8.
- Chu, A., Schoenberg, F. P., Bird, P., Jackson, D. D., and Kagan, Y. Y. Comparison of ETAS Parameter Estimates across Different Global Tectonic Zones. *Bulletin of the Seismological Society of America*, 101(5):2323–2339, Sept. 2011. doi: 10.1785/0120100115.
- Cox, S. and Scholz, C. On the formation and growth of faults: an experimental study. *Journal of Structural Geology*, 10(4):413–430, Jan. 1988. doi: 10.1016/0191-8141(88)90019-3.
- Drakopoulos, J. C. and Ekonomides, A. C. Aftershocks of February 19, 1968 earthquake in Northern Aegean Sea and related problems. *Pure and Applied Geophysics PAGEOPH*, 95(1):100–115, 1972. doi: 10.1007/bf00878858.
- England, P., Houseman, G., and Nocquet, J. Constraints from GPS measurements on the dynamics of deformation in Anatolia and the Aegean. *Journal of Geophysical Research: Solid Earth*, 121 (12):8888–8916, Dec. 2016. doi: 10.1002/2016jb013382.
- Ferentinos, G., Georgiou, N., Christodoulou, D., Geraga, M., and Papatheodorou, G. Propagation and termination of a strike slip fault in an extensional domain: The westward growth of the North Anatolian Fault into the Aegean Sea. *Tectonophysics*, 745: 183–195, Oct. 2018. doi: 10.1016/j.tecto.2018.08.003.
- Hainzl, S. and Ogata, Y. Detecting fluid signals in seismicity data through statistical earthquake modeling. *Journal of Geophysical Research: Solid Earth*, 110(B5), Mar. 2005. doi: 10.1029/2004jb003247.
- Hatzfeld, D., Ziazia, M., Kementzetzidou, D., Hatzidimitriou, P., Panagiotopoulos, D., Makropoulos, K., Papadimitriou, P., and Deschamps, A. Microseismicity and focal mechanisms at the western termination of the North Anatolian Fault and their implications for continental tectonics. *Geophysical Journal International*, 137(3):891–908, June 1999. doi: 10.1046/j.1365-246x.1999.00851.x.
- Heimann, S., Kriegerowski, M., Isken, M., Cesca, S., Daout, S., Grigoli, F., Juretzek, C., Megies, T., Nooshiri, N., Steinberg, A., Sudhaus, H., Vasyura-Bathke, H., Willey, T., and Dahm, T.

Pyrocko-An open-source seismology toolbox and library [Computer software. GFZ Data Services, 2017.

- Heimann, S., Isken, M., Kuhn, D., Sudhaus, H., Steinberg, A., Daout, S., Cesca, S., Vasyura-Bathke, H., and Dahm, T. Grond—A probabilistic earthquake source inversion framework V1.0 [Computer software. GFZ Data Services, 2018.
- Kapetanidis, V. and Kassaras, I. Contemporary crustal stress of the Greek region deduced from earthquake focal mechanisms. *Journal of Geodynamics*, 123:55–82, Jan. 2019. doi: 10.1016/j.jog.2018.11.004.
- Karabulut, H., Roumelioti, Z., Benetatos, C., Mutlu, A. K., Özalaybey, S., Aktar, M., and Kiratzi, A. A source study of the 6 July 2003 (Mw 5.7) earthquake sequence in the Gulf of Saros (Northern Aegean Sea): Seismological evidence for the western continuation of the Ganos fault. *Tectonophysics*, 412(3–4):195–216, Jan. 2006. doi: 10.1016/j.tecto.2005.09.009.
- Karakostas, B., Hatzidimitriou, P., Karakaisis, G., Papadimitriou, E., and Papazachos, B. Evidence for long-term precursors of strong earthquakes in the northernmost part of the Aegean Sea. *Earthquake Prediction Research*, 4:155–164, 1986.
- Karakostas, V., Papadimitriou, E., and Gospodinov, D. Modelling the 2013 North Aegean (Greece) seismic sequence: geometrical and frictional constraints, and aftershock probabilities. *Geophysical Journal International*, 197(1):525–541, Jan. 2014. doi: 10.1093/gji/ggt523.
- Karakostas, V., Tan, O., Kostoglou, A., Papadimitriou, E., and Bonatis, P. Seismotectonic implications of the 2020 Samos, Greece, Mw 7.0 mainshock based on high-resolution aftershock relocation and source slip model. *Acta Geophysica*, 69(3):979–996, Apr. 2021. doi: 10.1007/s11600-021-00580-y.
- Karakostas, V. G., Papadimitriou, E. E., Karakaisis, G. F., Papazachos, C. B., Scordilis, E. M., Vargemezis, G., and Aidona, E. The 2001 Skyros, Northern Aegean, Greece, earthquake sequence: Off—Fault aftershocks, tectonic implications, and seismicity triggering. *Geophysical Research Letters*, 30(1), Jan. 2003. doi: 10.1029/2002gl015814.
- Karamanos, C. K., Karakostas, G. V., Papadimitriou, E. E., and Sachpazi, M. Earthquake relocation in the wester termination of the North Aegean Trough. *Bulletin of the Geological Society of Greece*, 40(3):1091, June 2007. doi: 10.12681/bgsg.16828.
- Kim, Y. and Sanderson, D. J. Structural similarity and variety at the tips in a wide range of strike–slip faults: a review. *Terra Nova*, 18 (5):330–344, Sept. 2006. doi: 10.1111/j.1365-3121.2006.00697.x.
- Kim, Y.-S., Andrews, J. R., and Sanderson, D. J. Damage zones around strike-slip fault systems and strike-slip fault evolution, Crackington Haven, southwest England. *Geosciences Journal*, 4 (2):53–72, June 2000. doi: 10.1007/bf02910127.
- Kim, Y.-S., Peacock, D., and Sanderson, D. J. Mesoscale strikeslip faults and damage zones at Marsalforn, Gozo Island, Malta. *Journal of Structural Geology*, 25(5):793–812, May 2003. doi: 10.1016/s0191-8141(02)00200-6.
- Kim, Y.-S., Peacock, D. C., and Sanderson, D. J. Fault damage zones. Journal of Structural Geology, 26(3):503–517, Mar. 2004. doi: 10.1016/j.jsg.2003.08.002.
- Kiratzi, A., Tsakiroudi, E., Benetatos, C., and Karakaisis, G. The 24 May 2014 (Mw6.8) earthquake (North Aegean Trough): Spatiotemporal evolution, source and slip model from teleseismic data. *Physics and Chemistry of the Earth, Parts A/B/C*, 95:85–100, Oct. 2016. doi: 10.1016/j.pce.2016.08.003.
- Kiratzi, A. A., Wagner, G. S., and Langston, C. A. Source parameters of some large earthquakes in Northern Aegean determined by body waveform inversion. *Pure and Applied Geophysics PA-GEOPH*, 135(4):515–527, Apr. 1991. doi: 10.1007/bf01772403.

- Kissling, E., Ellsworth, W. L., Eberhart-Phillips, D., and Kradolfer, U. Initial reference models in local earthquake tomography. *Journal of Geophysical Research: Solid Earth*, 99(B10):19635–19646, Oct. 1994. doi: 10.1029/93jb03138.
- Klein, F. W. User's guide to HYPOINVERSE-2000, a Fortran program to solve for earthquake locations and magnitudes. 2002. doi: 10.3133/ofr02171.
- Konstantinou, K. Accurate relocation of seismicity along the North Aegean Trough and its relation to active tectonics. *Tectonophysics*, 717:372–382, Oct. 2017. doi: 10.1016/j.tecto.2017.08.021.
- Konstantinou, K. Estimation of optimum velocity model and precise earthquake locations in NE Aegean: Implications for seismotectonics and seismic hazard. *Journal of Geodynamics*, 121: 143–154, Nov. 2018. doi: 10.1016/j.jog.2018.07.005.
- Konstantinou, K. I., Mouslopoulou, V., Liang, W., Heidbach, O., Oncken, O., and Suppe, J. Present-day crustal stress field in Greece inferred from regional-scale damped inversion of earthquake focal mechanisms. *Journal of Geophysical Research: Solid Earth*, 122(1):506–523, Jan. 2017. doi: 10.1002/2016jb013272.
- Kourouklas, C., Papadimitriou, E., Tsaklidis, G., and Karakostas,
 V. Earthquake recurrence models and occurrence probabilities of strong earthquakes in the North Aegean Trough (Greece). Journal of Seismology, 22(5):1225–1246, June 2018. doi: 10.1007/s10950-018-9763-8.
- Kourouklas, C., Mangira, O., Iliopoulos, A., Chorozoglou, D., and Papadimitriou, E. A study of short-term spatiotemporal clustering features of Greek seismicity. *Journal of Seismology*, 24(3): 459–477, May 2020. doi: 10.1007/s10950-020-09928-1.
- Kourouklas, C., Console, R., Papadimitriou, E., Murru, M., and Karakostas, V. Modelling the large earthquakes recurrence times along the North Aegean Trough Fault Zone (Greece) with a physics-based simulator, Geophys. *Geophysical Journal International*, 225(3):2135–2156, Mar. 2021. doi: 10.1093/gji/ggab085.
- Kourouklas, C., Tsaklidis, G., Papadimitriou, E., and Karakostas,
 V. Analyzing the Correlations and the Statistical Distribution of Moderate to Large Earthquakes Interevent Times in Greece. *Applied Sciences*, 12(14):7041, July 2022. doi: 10.3390/app12147041.
- Llenos, A. L. and Michael, A. J. Modeling Earthquake Rate Changes in Oklahoma and Arkansas: Possible Signatures of Induced Seismicity. *Bulletin of the Seismological Society of America*, 103 (5):2850–2861, Sept. 2013. doi: 10.1785/0120130017.
- Lombardi, A. M. Estimation of the parameters of ETAS models by Simulated Annealing. *Scientific Reports*, 5(1), Feb. 2015. doi: 10.1038/srep08417.
- Lombardi, A. M. SEDA: A software package for the Statistical Earthquake Data Analysis. *Scientific Reports*, 7(1), Mar. 2017. doi: 10.1038/srep44171.
- Massey, F. J. The Kolmogorov-Smirnov Test for Goodness of Fit. Journal of the American Statistical Association, 46(253):68–78, Mar. 1951. doi: 10.1080/01621459.1951.10500769.
- McKenzie, D. Active Tectonics of the Mediterranean Region. *Geophysical Journal International*, 30(2):109–185, Dec. 1972. doi: 10.1111/j.1365-246x.1972.tb02351.x.
- Mesimeri, M., Kourouklas, C., Papadimitriou, E., Karakostas, V., and Kementzetzidou, D. Analysis of microseismicity associated with the 2017 seismic swarm near the Aegean coast of NW Turkey. *Acta Geophysica*, 66(4):479–495, May 2018. doi: 10.1007/s11600-018-0157-7.
- Nicholson, C., Seeber, L., Williams, P., and Sykes, L. R. Seismic evidence for conjugate slip and block rotation within the

San Andreas Fault System, southern California. *Tectonics*, 5(4): 629–648, Aug. 1986. doi: 10.1029/tc005i004p00629.

- Ogata, Y. Statistical Models for Earthquake Occurrences and Residual Analysis for Point Processes. *Journal of the American Statistical Association*, 83(401):9–27, Mar. 1988. doi: 10.1080/01621459.1988.10478560.
- Ogata, Y. Detection of precursory relative quiescence before great earthquakes through a statistical model. *Journal of Geophysical Research: Solid Earth*, 97(B13):19845–19871, Dec. 1992. doi: 10.1029/92jb00708.
- Ogata, Y. Space-Time Point-Process Models for Earthquake Occurrences. *Annals of the Institute of Statistical Mathematics*, 50(2): 379–402, June 1998. doi: 10.1023/a:1003403601725.
- Pacheco, J. F. and Sykes, L. R. Seismic moment catalog of large shallow earthquakes, 1900 to 1989. *Bulletin of the Seismological Society of America*, 82(3):1306–1349, June 1992. doi: 10.1785/b-ssa0820031306.
- Papanikolaou, D., Alexandri, M., and Nomikou, P. *Active faulting in the north Aegean basin*. Geological Society of America, 2006. doi: 10.1130/2006.2409(11).
- Papazachos, B. and Papazachou, C. *The earthquakes of Greece*. Ziti Publ. Co, 2003.
- Papazachos, B., Papadimitriou, E., Kiratzi, A., Papazachos, C., and Louvari, E. Fault plane solutions in the Aegean Sea and the surrounding area and their tectonic implication. *Bollettino Di Geofisica Teorica Ed Applicata*, 39:199–218, 1998.
- Papazachos, B. C. and Comninakis, P. E. Geophysical and tectonic features of the Aegean Arc. *Journal of Geophysical Research*, 76 (35):8517–8533, Dec. 1971. doi: 10.1029/jb076i035p08517.
- Papazachos, C. B. and Kiratzi, A. A. A detailed study of the active crustal deformation in the Aegean and surrounding area. *Tectonophysics*, 253(1–2):129–153, Mar. 1996. doi: 10.1016/0040-1951(95)00047-x.
- Petit, J. and Barquins, M. Can natural faults propagate under Mode II conditions? *Tectonics*, 7(6):1243–1256, Dec. 1988. doi: 10.1029/tc007i006p01243.
- Pichon, X. L. and Angelier, J. The hellenic arc and trench system: A key to the neotectonic evolution of the eastern mediterranean area. *Tectonophysics*, 60(1–2):1–42, Nov. 1979. doi: 10.1016/0040-1951(79)90131-8.
- Rocca, A., Karakaisis, G., Karacostas, B., Kiratzi, A., Scordilis, E., and Papazachos, B. Further Evidence on the Strike-slip Faulting of the Northern Aegean Trough Based on Properties of the August–November 1983 Seismic Sequence. *Bolletino Di Geofisica Teorica Ed Applicata*, XXVII(106):101–109, 1985.
- Rodriguez, M., Sakellariou, D., Gorini, C., Janin, A., D'Acremont, E., Pourhiet, L. L., Chamot-Rooke, N., Tsampouraki-Kraounaki, K., Morfis, I., Rousakis, G., Henry, P., Lurin, A., Delescluse, M., Briole, P., Rigo, A., Arsenikos, S., Bulois, C., Fernández-Blanco, D., Beniest, A., Grall, C., Chanier, F., Caroir, F., Dessa, J.-X., Oregioni, D., and Nercessian, A. Evolution of the North Anatolian Fault from a diffuse to a localized shear zone in the North Aegean Sea during the Plio-Pleistocene. *Geophysical Journal International*, 235(3): 2614–2639, Sept. 2023. doi: 10.1093/gji/ggad364.
- Sakellariou, D. and Tsampouraki-Kraounaki, K. *Plio-Quaternary Extension and Strike-Slip Tectonics in the Aegean*, page 339–374. Elsevier, 2019. doi: 10.1016/b978-0-12-812064-4.00014-1.
- Sakellariou, D., Rousakis, G., Vougioukalakis, G., Ioakim, C., Panagiotopoulos, I., Morfis, I., Zimianitis, E., Athanasoulis, K., Tsampouraki-Kraounaki, K., and Mpardis, D. DEFORMATION PATTERN IN THE WESTERN NORTH AEGEAN TROUGH: PRELIM-INARY RESULTS. *Bulletin of the Geological Society of Greece*, 50 (1):124, July 2017. doi: 10.12681/bgsg.11708.

- Thingbaijam, K. K. S., Martin Mai, P., and Goda, K. New empirical earthquake source-scaling laws. *Bulletin of the Seismological Society of America*, 107(5):2225–2246, Sept. 2017. doi: 10.1785/0120170017.
- Utsu, T. A Statistical Study on the Occurrence of Aftershocks. *The Geophysical Magazine*, 30:521–605, 1961.
- Wadati, K. On the Travel Time of Earthquake Waves. *Geophysical Magazine*, 7:101–111, 1933.
- Waldhauser, F. hypoDD-A Program to Compute Double-Difference Hypocenter Locations. 2001. doi: 10.3133/ofr01113.
- Waldhauser, F. and Ellsworth, W. A Double-Difference Earthquake Location Algorithm: Method and Application to the Northern Hayward Fault, California. *Bulletin of the Seismological Society of America*, 90(6):1353–1368, Dec. 2000. doi: 10.1785/0120000006.
- Wang, R. A simple orthonormalization method for stable and efficient computation of Green's functions. *Bulletin of the Seismological Society of America*, 89(3):733–741, June 1999. doi: 10.1785/bssa0890030733.
- Wells, D. L. and Coppersmith, K. J. New empirical relationships among magnitude, rupture length, rupture width, rupture area, and surface displacement. *Bulletin of the Seismological Society of America*, 84(4):974–1002, Aug. 1994. doi: 10.1785/bssa0840040974.
- Wessel, P., Smith, W. H., Scharroo, R., Luis, J., and Wobbe, F. Generic mapping tools: improved version released. *Eos, Transactions American Geophysical Union*, 94(45):409–410, 2013.
- Wiemer, S. and Wyss, M. Minimum Magnitude of Completeness in Earthquake Catalogs: Examples from Alaska, the Western United States, and Japan. *Bulletin of the Seismological Society* of America, 90(4):859–869, Aug. 2000. doi: 10.1785/0119990114.

The article Spatiotemporal characteristics and earthquake statistics of the 2020 and 2022 adjacent earthquake sequences in North Aegean Sea (Greece) © 2024 by P. Bonatis, is licensed under CC BY 4.0.